Author's response to reviews

Title: Detection of group A Streptococcus in tonsils from pediatric patients reveals high rate of asymptomatic streptococcal carriage.

Authors:

Amity L Roberts (amity.roberts@unmc.edu)
Kristie L Connolly (kconnoll@wfubmc.edu)
Daniel J Kirse (dkirse@wfubmc.edu)
Adele K Evans (akevans@wfubmc.edu)
Katherine A Poehling (kpoehlin@wfubmc.edu)
Timothy R Peters (kpoehlin@wfubmc.edu)
Sean D Reid (sreid@wfubmc.edu)

Version: 4 Date: 6 January 2012

Author's response to reviews:

To Whom it May Concern.

The typographical errors referred to under Major Revisions have been corrected as suggested. All minor changes have been made as well.

Please don't hesitate to contact me if additional changes are needed.